Author's response to reviews

Title: Unsuspected Pneumocystis pneumonia in an HIV-seronegative patient with untreated lung cancer: a case report

Authors:

Cai Chuang (skinblack1966@yahoo.com.cn)
Xie Zhanhong (Drxiezhanhong@163.com)
Gu Yinyin (DrGuyinyin@163.com)
Zhong Shuqing (Drzhongshuqing@163.com)
Zhong Nanshan (DrNanshanzhong@163.com)

Version: 3 Date: 10 July 2007

Author's response to reviews:

Dear Editor:

Following your advice and instructions, we have asked Professor Jeanne Bell, from University of Edinburgh to read and share her precious opinion on how to revise the manuscript.

We are glad to be able to resubmit the manuscript, we are sorry for the delay.

Best wishes!

Drs Chuang Cai, Zhanhong Xie, Yinyin Gu, Shuqing Zhong, and Nanshan Zhong

2007-7-10